Author's response to reviews

Title: Prediction models for short children born small for gestational age (SGA) covering the total growth phase. Analyses based on data from KIGS (Pfizer International Growth Database)

Authors:

Michael B Ranke (Michael.Ranke@med.uni-tuebingen.de)
Anders Lindberg (anders.lindberg@pfizer.com)
the KIGS International Board (Michael.Ranke@med.uni-tuebingen.de)

Version: 6 Date: 30 May 2011

Author's response to reviews:

Dear Editor

BioMed Central

Thank you for giving us a change to correct according your comments and the article is now revised accordingly. We hope that the MS is now suitable for publication.

Sincerely yours

Michael B Ranke